Author's response to reviews

Title: Clinical symptoms and chemotherapy completion in elderly patients with primarily diagnosed acute leukemia: a retrospective comparison study with a younger cohort

Authors:

Rong Hu (hurong1246@126.com)
Yong Wu (wuyong9195@126.com)
Xiaoying Jiang (vic_999@yeah.net)
Wenteng Zhang (w.t98@163.com)
Le Xu (xl_8332@163.com)

Version: 2 Date: 25 February 2011

Author's response to reviews:

Mr Jack Cochrane on behalf of Prof Dirk Vordermark
BMC Cancer.
e-mail: editorial@biomedcentral.com
Web: http://www.biomedcentral.com/
February 24, 2011

Dear Mr. Cochrane and Prof. Vordermark:

Re: MS: 8897457834217527
Clinical Symptoms and Chemotherapy Completion in Elderly Patients with Primarily Diagnosed Acute Leukemia: A Retrospective Comparison Study with a Younger Cohort

Thank you for the opportunity to revise the above referenced manuscript for publication in BMC Cancer. As requested, we are providing a Response Letter detailing revisions requested by the journal and Reviewers. We have incorporated all of the Reviewers’ suggestions into the revised manuscript and appreciate their very hard work in making our manuscript better. We look forward to its publication.

Sincerely,

Xiaoying Jiang, Professor
Nursing School of Fujian Medical University
#1 Xueyuan Road, Shangjie Town, Minhou County, Fuzhou 350108, Fujian, P.R. China
Tel: +86-138-6062-3495 Fax: +86-591-22862526
Email: jiangxy320@126.com